
All Exp Mtg
12-Jun-06MINOS Status Report

• Not bad for a Start-Up week
Intensity and rep-rate started low and slow, but both increased 
as the week progressed
A few tuning glitches

o Beam width at target a bit wider – will be worked on
o Beam hit baffle twice, which trips us off. Experts examining the

causes.
o Particle spray on Target worsens the existing water leak, which led 

to upgrades in He system – flow now about x2 more than in Feb.
Horn 1 ground fault early Monday AM

o Flake of crud? Pulsing the horn generally removes such oxidation
within a short time. Took longer this time. Back up as of ~noon.
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1.76 E18 
Integrated
protons on target
for the week

Getting ~1.2E13 per pulse
Pulse rep-rate slower than 
the nominal 2sec
But increased as the week 
progressed
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Detector up-time during beam

All data from last week 
at Target Position “150”
which we call “Medium-
High” energy. 
Accumulated 2 E18 POT.

Changed target position 
to High Energy during 
Sunday downtimes.
Record ~15 E18 POT
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Protons delivered per week
Total since May 2005    1.42 E20 protons

(1.399 E20 before June 1 2006


